Author's response to reviews

Title: Kidney disease in patients with type 2 diabetes: prevalence and associated variables in a random sample of 2,642 patients from a Mediterranean region (Catalonia, Spain).

Authors:

Gabriel Coll-de-Tuero (gabriel.coll@ias.scs.es)
Manel Mata-Cases (manuelmatacases@gmail.com)
Antonio Rodriguez-Poncelas (arpon@comg.cat)
Josep Mª Pepió (jmpepio@comt.org)
Pilar Roura (roura@cusido.com)
Josep Franch-Nadal (josep.franch@gmail.com)
Belen Benito (belen.beniro.badorrey@gmail.com)
Marc Saez (marc.saez@udg.edu)

Version: 2 Date: 29 March 2012

Author's response to reviews:

Dear editor,

according to your instructions I send a new version of the manuscript which have made the following changes:

a) a paragraph explaining the aspects related to informed consent and ethics committee approval

b) we deleted Figure 2 which corresponds to the flow-chart of the study. It has been incorporated as a Table

c) all tables have placed at the end of the manuscript, after the references.

Regards,

Antonio Rodriguez-Poncelas, MD
Research Unit
IAS, Salt, Girona, Spain.